

Received 9 May 2025, accepted 5 June 2025, date of publication 13 June 2025, date of current version 27 June 2025.

Digital Object Identifier 10.1109/ACCESS.2025.3579428

SURVEY

Unraveling the Smart Charging Technologies, Energy Sources, and Regulatory Standards for EVs



Department of Computing and Engineering, University of Huddersfield, HD1 3DH Huddersfield, U.K.

²Department of Artificial Intelligence, University of Management and Technology, Lahore 54000, Pakistan

³Department of Computer Science, COMSATS University Islamabad, Lahore Campus, Lahore 45550, Pakistan

⁴Department of Computer and Information Sciences, Northumbria University, NE1 8ST Newcastle, U.K.

⁵School of Computing Sciences, University of East Anglia, NR4 7TJ Norwich, U.K.

Corresponding author: Ansif Arooj (ansif.arooj@hud.ac.uk)

This work was supported in part by the Electric Vehicles Point Location Optimization via Vehicular Communications (EVOLVE), Marie Sklodowska-Curie Action (MSCA)-Research and Innovation Staff Exchange (SE), through Research Council (RC) under Grant EP/X039765/1 and Grant 101086218.

ABSTRACT Electric vehicles (EVs) are anticipated to be pivotal contributors to the global energy transformation in the automobile industry, ignited by their rapid proliferation. However, the widespread integration of EVs requires rigorous research and synthesis in charging solutions and EV supply equipment to meet the expected performance and improve ancillary services. Analysing the current landscape of EV charging technologies, it is essential to accelerate EV adoption by incorporating advanced control strategies that mitigate negative impacts and improve charging efficiency. Thus, this study presents a meticulous review that culminates in a curated anthology of 81 pivotal articles, with a predominant emphasis on charging EVs and associated technologies. The findings reveal extensive potential in this domain with our rigorous scrutiny of studies uniquely integrating all critical EV charging technologies, including their taxonomy, on-board and off-board charging infrastructures, and the global standards, and future research directions. The revelations of the study elucidate the prevailing paradigms within the EV research nexus, charting a course for future scholarly pursuits and practical applications.

INDEX TERMS Charging infrastructure and standards, regulatory frameworks for EVs, conductive and inductive charging.

I. INTRODUCTION

Electric vehicles (EVs) are transforming global transportation by offering a cleaner and more sustainable alternative compared to internal combustion engine (ICE) vehicles. The concept of EVs is not new: Between 1897 and 1900, they comprised approximately 28% of all automobiles, long before their modern resurgence [1]. However, the advent of ICE vehicles, powered by low-cost oil and technological

The associate editor coordinating the review of this manuscript and approving it for publication was Valerio De Santis^(D).

advancements, quickly dominated the market, causing EVs to be a historical footnote [2]. Today, with the harmful consequences of air pollution and environmental degradation becoming increasingly visible, EVs are emerging as a crucial answer to reach the Net Zero (NZ) goals [3], [4], [5]. According to the Environmental Protection Agency (EPA) [6], the transportation sector contributes significantly to greenhouse gas (GHG) emissions, contributing 29% of the total GHG emissions in 2022. Therefore, a massive shift to EVs as compared to ICE has the potential to drastically reduce this percentage. Breakthroughs in technology, such as

TABLE 1. Comparison with other studies.

Description	[17]	[18]	[19]	[20]	[21]	This Paper
The latest trends in charging technologies	\checkmark	\checkmark	\checkmark	Х	Х	\checkmark
Standards and Regulations for charging	Х	Х	√	Х	\checkmark	\checkmark
On-board charging resources	Х	Х	Х	~	\checkmark	\checkmark
Off-board charging resources	Х	Х	Х	Х	Х	\checkmark
Open issues and future prospects	~	~	~	√	Х	\checkmark

improved battery performance, charging infrastructure, and energy management systems, have further improved EVs widespread adoption.

EV charging systems can be evaluated using a variety of criteria, including battery charging technique, power flow guidance, usage of on- and off-board chargers, and power supply technology [7]. EV supply equipment (EVSE) is a critical component of a charging system that facilitates the power transfer between the EV and the grid [8]. EVs can be integrated into the grid using a combination of on- or off-board chargers that use alternating current (AC) or direct current (DC) [9]. Most commercial on-board chargers support unidirectional power flow, commonly known as grid-tovehicle (G2V), which is recommended due to its simplicity, reliability, affordability, and simple control technique [10]. Bidirectional chargers, on the other hand, offer vehicleto-grid (V2G) capabilities, which can return power to the utility grid [11], [12]. EV chargers incorporate sophisticated charging algorithms to allow optimal charging/discharge and adaptive power distribution by connecting to the EV and the electrical grid, thus improving the energy efficiency of the charger and reducing the load on the electrical grid [13]. The Institute of Electrical and Electronics Engineers (IEEE), the International Electrotechnical Commission (IEC), and the Society of Automotive Engineers (SAE) have recently greatly improved fast charging standards for AC and DC charging [14]. General accepted standards, along with regulatory regulations and standards, have been implemented for sustainable grid operation and to reduce negative consequences on the transmission network [15]. Rapid and ultra-quick charging stations attract attention since they are capable of charging the pack of batteries in a shorter time at higher levels of electricity [16]. Therefore, an exhaustive review of EV charging methods, on- and offboard couplers, their safety parameters, and standards is needed to prevent imminent problems and offer solutions.

A. PURPOSE OF THIS STUDY AND RESEARCH QUESTIONS

Over the years, extensive research has been conducted on various aspects of EV technology, covering topics such as battery advancements, charging infrastructure, energy management, and policy frameworks. However, the rapid evolution of EV technology, coupled with the urgent need for sustainable transportation solutions, necessitates a comprehensive and up-to-date survey that consolidates recent advancements and identifies emerging challenges. This study aims to bridge this gap by providing a state-of-the-art comparative analysis of existing EV research, highlighting key innovations, technological limitations, and future research directions. A summary comparing our findings with prior studies is presented in Table 1. Unlike previous surveys that focus on isolated aspects of EV technology, this work takes a holistic approach by examining both technical and infrastructural challenges, regulatory considerations, and the evolving role of EVs in smart grids and energy ecosystems.

- 1) What are the current charging methods used?
- 2) What sort of charging couplers are currently utilized in the EV infrastructure?
- 3) What are the various on-board energy sources?
- 4) What are the various off-board energy sources?
- 5) What are the standards for charging?

To the extent of our knowledge, this investigation represents a pioneering endeavour focused on developing an EV charging taxonomy and synthesizing the expansive framework encompassing various related facets.

The paper is structured as follows: Section II presents the descriptive analysis of EV charging methods to address the answers to 1. In this section three charging methods: conductive charging II-A, inductive charging II-B, and battery swapping II-C are discussed in details. Section III covers the currently used charging couplers for EV infrastructure. Conductive and inductive charging couplers are studied in III-A and III-B, respectively. Section IV covers available on-board energy sources focused on different batteries, capacitors, and available cells. Section V covers seven different types of off-board energy resources. Section VI provides a comprehensive review of research questions pertaining to the EV charging standards and regulations. EV charging technology standards are covered in section VI-A, followed by the coupler/connector standards in section VI-B and finally the communication standards in section VI-C, respectively. Finally, a general discussion of future work is presented in sections VII, followed by a conclusion in section VIII. The taxonomy of this study is illustrated in Fig. 1.

II. EV CHARGING METHODS

An EV charger is a device that converts the power from an external source into an energy form suitable for the battery system [14]. There are typically three primary ways to charge EVs: (i) conductive charging, (ii) inductive charging, and (iii) battery swapping. As conductive charging is much more affordable and effective, charging station operators prefer it [22]. However, several development



FIGURE 1. A comprehensive taxonomy of EV charging methods, couplers, energy sources, and standards: A visual guide to smart charging technologies.



FIGURE 2. Levels and modes of EV charging.

initiatives are enhancing inductive charging, as it improves

user convenience [19], [23]. For battery swapping to be

conceivable, standardized battery measurements and internal

connections are required, and batteries from every manu-

facturer would have to be almost the same in terms of

dimensions, form, and kind [24]. This severely restricts

the freedom of design and narrows the battery placement

A. CONDUCTIVE CHARGING

Conductive charging systems utilize charging cables and connectors between the EV connector and the internal charge unit [21]. The EV's battery receives electrical power from the charging point through these connectors [19], [21]. There are three different charging levels and four different modes of charging as shown in Fig. 2. Let us discuss them in detail in the relevant sections.

options.

Parameters	Level 1	Level 2	Level 3	
Current	AC	AC	DC	
Voltage	120 V	208V-240V	400-1000V	
Charging Power	1.3-2.4kW	2.9-19.2kW	50-350kW	
Time (EV)	40-50 hours	4-10 hours	20 min - 1 hour	
Time (Hybrid)	5-6 hours	1-2 hours	N/A	
Power Source	Wall outlet	In-house and	Public charger	
I Ower Source	wan outlet	Public charger	i done charger	
Connector Types	SAE J1772	SAE J1772, Tesla	CCS, CHAdeMO, Tesla	
Grid Voltage	120 VAC, 230 VAC	240 VAC, 400 VAC	208-600 VAC /VAD	
Power Limits	\leq 3.3 kW	3.3 kW - 20kW	>50 kW	
Issues	Slow charging, limited	Moderate charging, installation	Rapid charging, costly infrastructure, battery	
188008	power, home-use only	costs, grid load, charging availability	degradation, inconsistent compatibility, grid stress	
Application	Home, Offices	Home, Offices, Public	Public Only	

TABLE 2. Summary of EV charging levels [18], [20], [25], [27], [28], [29], [30], [32], [33].

1) LEVELS

- Level-1 on-board: This technique is also known as *home charging*. A grounded electrical outlet and a proper cord set to transfer current via an internal charger is used. This level supports connection to the existing outlets [20], [25], [26]. However, it uses a single-phase AC charging approach that integrates the motor and inverter within a dual-inverter system [27]. This method aims to enhance the voltage state of the AC charging compared to conventional chargers.
- Level-2 on-board: This charging point charges the EV rapidly than a Level-1 charger. An EV must be equipped with an on-board charger capable of using an AC EVSE for power. The term "fast AC charging" denotes charging at levels of 7kW (32A single-phase) or 21kW (three-phase). The main EV charging technique involves extending AC electricity from the electrical source to power an EVSE-specific on-board charger. Most significantly, home, office, and public charging stations can utilize this level [20], [28]. This specific charge level is used where a single economically priced charger can serve multiple residential buildings or families. A two-level optimal hybrid EV charging approach by adaptive-equivalent consumption minimization and dynamic programming was developed in [29].
- Level-3 off-board: This is also referred to as DC rapid charging. It provides high-power DC directly to the vehicle's battery, allowing for much quicker charging in comparison to Level-1 and Level-2 connectors. Level-3 chargers are often found in public or commercial locations and can quickly charge an EV to 80% in about 20-30 minutes, making them ideal for long journeys and emergency charging [28]. A suitable offboard charger can provide energy to the conductive charging system architecture. The DC power source can vary from AC Levels-1 and -2 to extremely high power levels, capable of recharging an EV battery to over 50% in as little as 10 minutes. There is a wide range of power levels at which an EV battery can be charged [30]. To accomplish rapid DC fast charging, Levinson et al. [31], [32] introduced an algorithm for

an off-board converter using three phases to regulate the maximum power level electricity grid from an exaggeration induced by the heightened demand for EV fast charging. Nonetheless, apartment building occupants can efficiently utilize Level-2 charging at a minimal cost of charging [33].

Finally, Table 2 presents summary of charging levels.

2) MODES

The charging method specifies the protocol for a safe interlink between the EV and the charging facility. Globally, these standards are uniform [34]. A potential method to recharge an EV is to connect the on-board charger via the main AC supply system. Using an off-board charger to deliver a DC supply is an additional EV charging technique. Fast charging could be achieved by using a high-power charging infrastructure [35].

- Mode-1: It is commonly utilized for home charging [34], [35]. The EV is connected to the AC power supply by employing the standardized electrical sockets at the supply edge that use both power and protective ground conductors. To operate in this mode, the electrical configuration must adhere to safety requirements and possess an earth leakage protection system. An electrical circuit breaker of 10-A and sockets are essential for the earthing system to provide overload protection and prevent accidental contact [33], [35], [36], [37].
- Mode-2: The EV is connected to the electrical grid through common household outlets. To recharge, an earthing cable must be installed along with a network, that is either single or three separate phases. There is an internal safety feature on the cable. This particular method is costly as compared to mode-1 [36], [37], [38]. Mode-2 charging is accomplished within the home using a common power outlet and an additional cable EVSE, frequently referred to as an *occasional cable*, which typically comes with an EV from its maker. The in-cable control box or in-cable control and protection device (IC-CPD), which has an integrated current limit, is the "box" on the cable. Similar to a charge point, the IC-CPD can exchange data with the EV and has

protection inside of it that would not be present in mode-1. The maximum charging current for a typical domestic outlet is 10A. This restriction to 10A leads to an aggregate charging rating of 2.3kW [37]. Overheating and potential fire could occur from a frequently used socket if the charging current is not restricted to 10A [39], [40], [41].

- Mode-3: Through a designated socket, plug, and circuit, the EV is directly attached to the infrastructure. Surveillance and protection features are reinstated in the installation permanently. This is the only charging option that conforms to the regulations governing electrical connections. It enables load-shedding, which extends the duration of EVs that can be charged along with the operation of electrical household appliances [38], [41]. It has a higher power output compared to mode-2. In this mode of charging, a control circuit extends to operate devices, and the EV remains permanently linked to the AC power supply via a specialized EVSE [24].
- Mode-4: An external charger is used to connect the EV to the primary electrical infrastructure. The vehicle charging cable and control and protection components have been permanently embedded in the integration [31], [40]. Moreover, in this mode, the DC charging station is able to be used for home or public charging. The EV charger is a component of the charging point, rather than the EV itself [24]. During this charging mode, an off-board charger connects the EV to the AC network, and the pilot control operation applies to equipment that is connected to the AC supply [42].

B. INDUCTIVE CHARGING

This is a new technology that transfers power from a charging facility to an EV without the use of physical connectors. It operates via electromagnetic transmission and is also referred to as inductive power transfer (IPT), wireless power transfer (WPT), contactless power supply (CPS), or contactless energy transfer (CET) [43]. The electromagnetic field produced by the swapping current affects the EV's receiver, which is also made of wires, causing an induced current to pass through it to replenish the battery. This system comprises a setup of a charging pad, which serves as the main component placed at the power outlet, and a receiving pad, which is the secondary component placed underneath the EV. Water and non-metallic substances such as flagstone, asphalt, and other materials are used for energy transfer [18], [28], [44]. This unique method carries power over an inadequate air gap, resulting in a frictionless charging experience [38].

1) TYPES

Understanding these types is critical because they influence the prospects of electric transportation, increase user convenience, and promote universal adoption [44], [45]. Each type is designed to meet certain charging requirements and operational conditions.

- Resonant Inductive Wireless Charging (RIWC): RIWC is a revolutionary technique employing magnetic resonance technology. In contrast to conventional inductive charging, which depends on the charger and receiver being close to one another, RIWC functions more efficiently at longer distances [43], [45]. Resonant inductive coupling, in which both the transmitting as well as receiving coils are set to a certain resonant frequency, is used to accomplish optimal energy. This increases the transmission of energy over an air gap. The technology makes it possible to charge contactless without requiring physical connectors, which is especially useful for open spaces and urban settings where accessibility and usability are crucial. Roadways, driveways, and garages may all be equipped with RIWC systems, which enable EVs to recharge while parked or moving. RIWC offers a more flexible and user-friendly charging solution in EVs, thus has the better potential for development [46].
- Inductive Wireless Charging (IWC): In IWC wireless energy transfer occurs when the AC passes via the primary coil, producing a magnetic flux that oscillates and enables an electrical current that circulates through the secondary coil. The 50kW CET system for charging large EV batteries was developed in [46]. A hybrid system that combines the propulsion system and the CET system is currently under development. Despite early reservations, this technology offers exceptional energy transfer efficiency even in larger air gaps between the transmitter and the EV load receiver [46], [47]. A secondary induction coil within the mobile device records energy from the field having electromagnetic parameters and converts it again into electrical power to recharge the battery pack. Induction chargers function by utilizing an induction coil embedded in a charging base to generate an alternating electromagnetic field. The proximity of the two induction coils forms an electrical transformer. Resonant inductive coupling in the charging system enables greater distances between the sender and receiver coils. To reduce the weight and resistance caused by the skin effect, this resonant system has been improved by adding a movable transmission coil and using alternative materials for the receiver coil, such as aluminum or silver-plated copper [48].
- Capacitive Wireless Charging (CWC): By using electric fields instead of magnetic ones, CWC, is a technique for transmitting electrical power between two objects. A pair of conductive plates is used in this technique: a transmitter plate that is attached to an electrical source and a receiving plate that is attached to the item that needs to be charged. Energy is transferred wirelessly when an AC is delivered to the transmitter plates [47], [48]. This causes an electrical field that oscillates, causing a displacement current to be induced

in the receiver plates. CWC has several benefits over inductive techniques, including the capacity to recharge via non-metallic materials such as plastic or glass and less electromagnetic interference. It is especially helpful in applications where seamless design and limited space are essential. For effective energy transfer, CWC usually needs exact alignment and near-distance transmission device and receiver plates. The effectiveness of CWC is improved by development in materials and circuit design, which will result in a viable technology for the upcoming wireless charging applications [5], [45], [49].

2) MODES

There are three charging modes: stationary, dynamic, and semi-dynamic. Each mode uses electromagnetic induction to WPT, although their diversity depends upon EV position, charging time, and operating conditions. EVs can use one of three WPT modes designed to accommodate a variety of charging conditions and mobility needs.

- Stationary Wireless Charging (SWC): SWC refers to the method of charging an EV while it is parked and inactive for an extended period of time at a fixed location. SWC's technology is evolving [43]. With SWC, charging is much more secure and convenient. With regard to vehicle operation, charging station assignment, charging time and frequency, SWC and conventional plug-in conduction charging are identical. Because their batteries have a short range, EVs require additional charging sessions to travel farther [45], [46]. Technologies that store and accumulate energy for later use are known as energy storage systems. Rechargeable batteries, an electronic management system, an inverter, and a thermal framework are the primary components of an SWC. An option for using the stored energy is to power a particular facility or an electrical grid. Batteries of a stationary repository are coupled to a renewable energy generator, such as solar or wind power.
- Dynamic Wireless Charging (DWC): The term DWC refers to the concept of wirelessly charging an EV when in motion. This concept was originally developed at the University of California, Berkeley [50]. Researchers from the Korea Advanced Institute of Science and Technology created the initial release for the sale of dynamic wireless charging technology, called online (OL)EV [51], at the beginning of 2009. The receiving component in DWC is positioned atop the EV body frame, and the transmission components are mounted on a minuscule pavement patch to supply charging for the moving EVs. The assortment of inductive wires or circuits beneath the road surface provides power to the vehicles that use the system [38], [52], [53]. Due to high costs, the technology's commercialization attempts have failed due to its low efficiency [54].
- Quasi-Dynamic Wireless Charging (QDWC): The term QDWC refers to a charging method that incorporates

some modification of the dynamic or real-time charging parameters to optimize the charging process based on variables such as, for example, the conditions of the grid, the price of electricity or other pertinent variables [55]. It is categorized as either fully dynamic or real-time charging, where parameters are continually modified based on changing conditions, or static charging, where parameters remain static through the charging session. In QDWC, to reconcile the EV owner's needs, the power grid, and potentially other stakeholders, the charging system may periodically evaluate outside variables and perform adjustments to different parameters such as charging rate or voltage. This strategy seeks to optimize the charging process's effectiveness, dependability, and affordability. One potential solution in a quasi-dynamic charging system would be to factor in the daily fluctuations in electricity prices and modify the charging rate accordingly, allowing EV owners to take advantage of cheaper periods [56]. As an alternative, it might react to grid conditions and modify the charging rate to prevent overloading the power grid at times of high demand [57]. To achieve an equilibrium between simplicity and productivity, the QDWC concept allows for some degree of flexibility and optimization in EV charging without the ongoing adjustments seen in fully dynamic systems [55].

C. BATTERY SWAPPING

The third alternative involves swapping out depleted batteries for substitutes at an exchange station. Battery swapping also known as battery switching, is a technology used in EVs that enables them to quickly swap out a fully charged new battery pack for a discharged one, saving time compared to charging the vehicle at a designated point [58]. The potential of battery swapping to transform electric transportation and offer a workable and expandable solution for diverse automotive market segments is highlighted by its versatility across different vehicle types [59], [60]. Battery swapping is used frequently in applications involving Forklifts that are powered by electricity [59]. With about 11,000 GoStations in Taiwan and 250 in Mainland China, The largest network for electric moped battery swapping is presently operated by the Taiwanese electric scooter company Gogoro [61]. The Chinese luxury manufacturer Nio is the only noteworthy operator of automobile battery-changing facilities for consumers. The procedure takes three minutes from beginning to end, and the company has developed about 2250 battery-change units throughout Europe and China [59]. Initiatives were made by Tesla and Renault as well. Generally, battery swapping will depend upon the station, battery type, end-user, or service provider. These techniques are explained below:

- By Station Type:
 - Manual Swapping: This method replaces the battery manually by individuals or technicians rather than

Region	Connector	Pins	Protocol	Power	Levels	Range/Hour	Outlet	Image
North America	SAE J1772 (Type 1)	5	PWM	19.2 kW (AC)	Level 1, Level 2	10-50 miles	L120V (L1), 240V (L2)	Type 1 5AE J1772 / IEC 62196-2
North America	CCS1 (Combo 1)	7	PLC	350 kW (DC)	Level 3, (DCFC)	180-240+ miles	400V-, 1000V DC	CCS Combo 1
Europe	Type 2 (Mennekes)	7	PWM	43 kW (AC)	Level 2, Level 3	20-75 miles	240V (L2), 400V (L3)	Type 2
Europe	CCS2 (Combo 2)	9	PLC	350 kW (DC)	Level 3, (DCFC)	180-240+ miles	400V-, 1000V DC	CCS Combo 2
Japan	CHAdeMO	10	CAN	400 kW (DC)	Level 3, (DCFC)	180-240+ miles	400V-, 1000V DC	CHAMMO
China	GB/T AC	7	CAN	43.5kW (AC)	Level 2	20-75 miles	240V, 400V AC	GB/T AC GB/T 202034.2
China	GB/T DC	9	CAN	4900 kW (DC)	Level 3, (DCFC)	200-300+ miles	400V-, 1000V DC	GB/T DC
Tesla (Global)	Tesla (NACS)	5	Proprietary	250kW (DC)	Level 2, Level 3,	150-250 DC	240V (L2), 400V- 1000V DC	

utilizing sophisticated machinery. When an owner or driver brings their EV to the station, technicians manually remove the discharged power battery and change it out for one that is completely charged. During this procedure, it might be necessary to secure the replacement battery, carry out quality inspections, and guarantee correct installation [22]. Compared to traditional charging methods, the EV owner can drive again after the swapping is finished because the power unit is completely charged. Although it was investigated in the initial stages of the formation of EVs, manual battery swapping has encountered difficulties. As a result, automated or semi-automated solutions have become more popular to address effectiveness as well as security challenges.

- Automatic Swapping: In this method, the charging stations use a robotic system to deactivate the EV's depleted battery and change it out for a fully powered one. Immediately as the EV pulls into a designated station, an automated system finds and replaces the battery with a fully charged one [59].
- By Battery Type: Techniques for battery swapping can use a variety of battery classifications, for instance,

hybrid nickel-metal and lithium-ion batteries. At a battery swapping station, a fully charged lithium-ion or nickel-metal hybrid battery is swapped out for a depleted one. The versatility of the switching strategy in supporting various battery compositions promotes the growth of EV technologies [22], [62].

- By End User: For EVs, battery swapping is a flexible solution that can be used by end users of both passenger and commercial vehicles. This strategy is especially pertinent to passenger cars, where easy and quick recharging is essential to improving the user experience and promoting the general use of electric mobility. Fleet managers can continue operating their vehicles continuously by using battery swapping as a time-effective substitute for extended charging times for commercial vehicles, such as electric delivery vehicles and buses.
- By Service Type: EV battery swapping services can be designed using a variety of service models, such as pay-per-use and subscription. Under a subscriptionbased model, customers pay a set amount each month to receive access to a predefined number of battery swaps over a predetermined period. This strategy fits the needs of consumers who value a set, recurring

cost for the maintenance of their EVs by providing predictability and convenience. Conversely, a pay-peruse model offers flexibility to users who might not need to replace their batteries frequently by charging them only when they swap them out. This payment model is appropriate for users who have different usage patterns or who would rather pay for the service in more discrete ways [22], [61].

III. INFRASTRUCTURE FOR CHARGING COUPLERS

Various EV charging connectors and couplers are available, each with special features and functionalities. Significant variations in the power rates, standards, and configurations of EV chargers and couplers are present in different regions. To prevent disputes and difficulties, manufacturers and regulatory organizations are working to ensure compatibility by developing universal standards and protocols [40], [63]. The summary of region-wise classification of the charging inlets is shown in Table 3. EV battery holds DC energy, grid power is provided as an AC electricity. Therefore, a conversion from AC to DC is required, which is achieved by the EV onboard charger. For DC fast charging, the charging station is required to convert AC to DC before being utilized by the EV. Because DC fast charging provides direct power to the battery than through the vehicle's onboard charger, it enables a faster charging solution [15], [19], [26], [32]. In this section, the charging connectors and couplers that are currently utilized in the EV infrastructure will be covered.

A. CONDUCTIVE CHARGING COUPLERS

The main component of is the electrodes and can be further classified into two main systems, described hereunder:

1) PLUG-IN CHARGING CONNECTORS

There are six classifications of plug-in charging connectors, each designed to operate with a particular charging standard and power level. Each connector has a unique design, compatibility, and charging speed to meet regional needs.

1) SAE J1772: It is a standard EV connector extensively used in North America, offering a flexible and reliable option. It works with both Level-1 and Level-2 charging stations and comes with a single-phase electrical charging mechanism. To guarantee safe charging, the connector is designed to include safety parameters such as a ground pin and a locking framework. In May 2011, its parameters were updated to align with the IEC Type-1 coupler or IEC 62196 – 2 standard. Since AC battery charging systems are compatible with power sources and have a 19.2kW charging capability, they are widely used in the US, Canada, and Japan. This coupler features five pins in three different sizes and a circular diameter of 43.8mm [64]. The EV's control system can limit mobility while connected to the EVSE and notify the EV of the latch push key by using the proximity pilot detecting pin. One type of communication signal is through the control pilot pin. It is used to transmit additional data as well as the charging level information across the EV environment. The transmitted signal is employed to control the start/stop charging process, convey the permitted charging current limits, and detect the presence of an EV [26], [64].

- 2) IEC Type-2: The German business Mennekes introduced the Type-2 connector in 2009, typically referred to as the Mennekes Type-2. Its status as the industry standard for charging EVs has been cemented by its extensive adoption throughout Europe. The European Commission was recognized as a regulated charging coupler in January 2013 [65]. The same coupler is being used by Australia and New Zealand, two non-EU nations, while a comparable but distinct design is explained as Guobiao (GB)/T 20234 - 2. The IEC 62196-2 states that the original purpose of this coupler was intended for use with AC charging solutions. That indicates that it can be charged by an AC power supply that is either single-phase or three-phase of up to 7.4kW or 43kW, respectively. This type-2 coupler has three possible configurations [66], [67]. Furthermore, the type-2 IEC connection features a smooth side and a cylindrical 70mm size for use with mechanical alignment. Certain public EV charge outlets in Europe, Australia, New Zealand, and various other locations have socket outlets solely, whereas others provide charging cables. Additionally, a portable EVSE is also provided. This EV charging cable features a Schuko socket and a device for IC-CPD, enabling AC charging through a standard home outlet. It was designed in collaboration with Siemens and Mennekes [64].
- 3) Combined Charging System (CCS) Combo: Europe has access to DC fast charging, as in North America, with the CCS Combination 2 standard. Similarly to the J1772 connector of the CCS Combo-1, it combines the Type-2 connector with two DC rapid charging outlets [64]. The CCS Combo specification uses a single EV input to manage both AC and DC fast charging levels.
- 4) CHAdeMO: It merges dual DC-fast battery ports in addition to the Type-2 connector, much comparable to the CCS Combo 1 (J1772-2) across the United States and Canada [26], [66]. In 2021 and 2022, respectively, the complete versions of ChaoJi-2, (CHAdeMO-3.0.1) and ChaoJi-3 (CHAdeMO 3.0) were made public. Significantly, the IEC has approved the newly proposed specification 61851 23; 62196 3; and 62196 3 1 for the ChaoJi charging mechanism and coupler. In 2023, field testing of ChaoJi-2 (CHAdeMO-3.0.1) was combined with the deployment of Ultra-ChaoJi (CHAdeMO-3.0.1) and Ultra-ChaoJi (CHAdeMO-4.0) in Japan [40].
- 5) GB/T: A collection of national standards for connections used in a variety of industries, especially communications, electronics, and EVs, is commonly referred

to as the GB or T standard connector. The GB/T series, which was established by the Chinese Standardization Administration, specifies the mechanical, electrical, and physical characteristics of connections to guarantee safety and interoperability. The GB/T-20234 standard specifies connectors for EV charging and includes various voltage and current requirements [68]. The China electrical charging standard (GB/T 20234.2 – 2015) defines cables having Type-2 male plugs at both terminals and female plugs on EVs, as well as unique control signaling, which is different from the rest of the world [19]. In addition, the third part of the GB/T-20234 standard contains the fundamental prerequisites, functional definitions, type components, features, and parameters for EV charging. GB/T-20234.3 - 2015 contains couplers working in charging mode-4 and connection mode C, with maximum voltages and peak currents of 1000V DC and 250A DC, respectively [66]. With the support of the governments of China and Japan, the CHAdeMO Organisation and CEC worked on developing a new global retroactive DC electrical charging standard that is fully compatible with the existing GB/T and CHAdeMO standards. At the CHAdeMO summit, a new GB/T-CHAdeMO coupler prototype was unveiled with a charging capacity of as much as 900 kW at 1500 V [19]. ChaoJi is the formal term for the forthcoming standard for EV chargers. ChaoJi standard will interface with EV charging devices over the CAN, comparable to its predecessors. This collaborative effort would culminate in a single, harmonized standard [15], [16], [26].

6) Tesla: This coupler is a unique charging interface made exclusively for Tesla EVs and is also referred to as the Tesla charging connector. The minuscule and sleek design of Tesla's connector, which combines AC and DC plugs into a sole socket, is widely recognized. The Tesla-developed connector is used by Tesla automobiles in North America for both rapid charging DC and at-home charging AC. Other parts of the world, notably Europe as a whole, have Tesla vehicles employing the more popular CCS plug [26]. The company's extensive charging network includes the Tesla EV coupler, which provides simple but efficient energy transmission for Tesla EVs. The firm now offers four EV models in its lineup, after the 2008 Roadster as its first: the Roadster 2020, Model-S, Model-3, Model-X, and Model-Y [69]. As a result, the US variant is also offered for sale in Mexico, Canada, Taiwan, and Japan, and has the exclusive Tesla inlet. Moreover, the Tesla offered for sale in Europe, on the other hand, is equipped with a Type-2 port or, more advanced, a CCS Combo-2 connector. Lastly, China has a twin inlet described as AC GB/T and DC GB/T inlets. With the installation of its own EV charging stations throughout the globe, often referred to as Tesla Superchargers or Tesla Megachargers thus the Tesla standard is gaining traction. Moreover, the 2.4kW, 120V trickle, 19.2KW, and 144kW 480V rapid charging rates are offered by these EV charging stations [69]. With a single connection, all three of these levels and frequencies are possible, enabling driving ranges of 3 to 273km/h of charge [70].

2) AUTOMATED AND HEAVY DUTY EV CHARGING CONNECTORS

These automated and heavy-duty EVs (HEV) can be divided into three types of charging connectors. The details are as follows:

- 1) Charging Robot Coupler: The most sophisticated and popular charging technique for HEVs is the charging cable method, which is the foundation of the passive conductive charging system that uses charging automation. This system requires a transformer and a rectifier in addition to the conductor, which can be placed inside or outside the EV. The HEV can be charged using DC, AC, or both simultaneously, depending on the setup. Currently, 500kW DC and 43kW AC of charging power are achievable. Even greater charging power over a megawatt will be possible by using the megawatt charging system that is presently being developed [71]. These couplers use robots that are outfitted with camera-based recognition to align and connect charging plugs. These robots use data about the particular vehicle, for instance, the precise location of the outlet for charging and the total amount of electricity required. The charging robot uses this information to align the charging connector with the charging socket, insert it into the connector, and then take it out of the socket. The accuracy of the robot's vision, control, and mobility systems, as well as the tolerance ranges of its components and the surrounding temperature, vibrations, and environmental factors, all affect its accuracy. Prominent firms in this domain comprise ROCSYS, Siemens, and Stäubli, all of which have made noteworthy progress. Stäubli's quick charging connector system delivers more than a MW of power for demanding conditions, Siemens' prototype gives up to 300kW of energy for charging tested with EINRIDE, and ROCSYS' robots are utilized by numerous businesses and at public facilities in the USA [19].
- 2) Under Body Coupler: A static or movable vehicle as well as a stationary or movable ground component, makes up an underbody-coupling system. The vehicle segment is fastened with the vehicle's underside, while the ground segment is integrated into or set on top of the charging point. The ground and the vehicle segment need to line up to allow the vehicle to be charged. For charging systems, the maximum amount of misalignment is measured at ± 150 mm off the ideal

parking placement over the axes of the car [24]. Assistance systems improve parking precision, and with the help of particular interface architectures and guidance systems, the electrical interfaces need to be physically joined once they are aligned. A voltage regulator, transformer, transmission system, and control system are also needed for the charging technology. Under the AC/DC Project, Volterio's underbody-coupling system is being tested in Graz, Austria for electric passenger vehicles. It has attained Laboratory Testing of Integrated and Semi-Integrated Systems (TRL-5), signifying validation in a relevant context. TRL-5 is also possessed by comparable systems from Fraunhofer IVI as well as Schunk Transit Systems, which were tested in the AULA Project and will be showcased in the Mega-Laden Initiative for HEVs. On the other hand, Elonroad's system achieved TRL-7 after being evaluated by an end user in a working context [72].

3) Overhead Contact Line (OCL): Electric buses and trucks are the main applications for the OCL EV charging system, which involves charging the vehicles using overhead power lines while moving or at predetermined pauses. Pantographs or other such devices installed on the vehicle's roof and connected to overhead electric cables are used in this manner to extract electricity. Similar to conventional trolley bus or tram systems, the system's continuous power supply negates the condition for sizable onboard batteries, thereby enabling greater operating ranges between recharge stations. Three pilot projects titled, "Field Test E-Highway Schleswig-Holstein", "eWay Baden-Württemberg", and "Electrified Innovative Heavy-Duty Transport on Autobahns" were developed in Schleswig-Holstein, Baden-Württemberg, and Hessen, respectively. These projects evaluate the viability of electric heavy-duty trucks that directly take electricity while driving via OCL. These lines are connected to trucks with pantographs, allowing for continuous charging and lowering dependency on fossil fuels. Every project assesses this technology's effectiveness, scalability, and performance in real-world scenarios across many geographies and terrains. Furthermore, Siemens developed the technology that is used in these projects. A pantograph is a component of this technology, can connect or disengage with the OCL reaching up to 90km/h, and will cut off on its own if the vehicle swerves or indicates [24]. High efficiency and the capacity to accommodate heavy-duty electric vehicles on lengthy routes with little downtime are the primary benefits of overhead contact line (OCL) charging. Because infrastructure may be incorporated into existing bus routes or dedicated lanes, it is especially advantageous in urban locations with well-established public transit systems. However, putting in overhead lines can be expensive and obtrusive, requiring a large investment and urban planning. Nevertheless, OCL EV

charging is a practical and effective way for cities to electrify their fleets of public and commercial transportation while also lowering emissions and improving the sustainability of public transportation [19].

B. INDUCTIVE CHARGING COUPLERS

Portable electronics are powered by electromagnetic induction. The portable device does not have to line up exactly or establish an electrical connection with a terminal or connector positioned adjacent to a charging station or inductive pad. EVs require an onboard contactless charger, which adds weight to the vehicle. The different charging couplers are:

- SAE J2954: The SAE designed AE J2954 as a wireless charging standard. It outlines the specifications for WPT systems, guaranteeing safety and interoperability with different manufacturers. The standard concentrates on effective and dependable charging with little electromagnetic interference and supports power levels up to 11kW. To enhance user experience to promote the broader use of electric mobility, SAE J2954 seeks to facilitate simple and convenient charging of EVs eliminating the requirement for physical connectors [40], [74].
- 2) Wireless Power Consortium (WPC): The WPC designed the Qi standard, which has been extensively utilized for wireless charging in commercial gadgets, but it may also be used for EV applications. The goal of WPC's EV charging initiatives was to provide a single, worldwide scalable infrastructure. The goal of the Qi norm for EVs is to make wireless charging simple and effective by utilizing the knowledge gathered from numerous millions of Qi-enabled consumer electronics. The strategy employed by WPC guarantees that EV charging may be widely available and user-friendly, which encourages the expansion of EVs [75], [76].
- 3) ChaoJi Wireless: With an emphasis on high-power wireless charging options for electric cars, ChaoJi Wireless is an emerging specification that was created mostly in Asia. It is made to meet the demands for instance buses and trucks by supporting power levels that are far greater than the present norms. To facilitate quicker adoption and more flexibility in the general public and corporate transportation sectors, ChaoJi Wireless seeks to offer a reliable and effective recharging service that can meet the demanding power needs of commercial EVs [77].
- 4) WiTricity Halo: WiTricity Corporation created the exclusive WiTricity Halo wireless charging system, which is renowned for its cutting-edge resonant inductive coupling technique. Possessing a maximum potential of 11 kW, the Halo system is intended to provide effective and powerful mobile charging for EVs. Convenience and compatibility are prioritized by WiTricity's technology, enabling smooth

integration across various car models and manufacturers. WiTricity Halo seeks to promote sustainable mobility and encourage the broad adoption of EVs by concentrating on intuitive and effective wireless charging solutions [78].

5) Plug-less Power (Evatran): Evatran created Plug-less Power, one of the initial wireless EV chargers to sell commercially. It allows EV owners to charge their vehicles hands-free using inductive charging technology. Power transmission levels range from 3.2 to 7.2 kW, the system is appropriate for commercial and residential uses. By rendering the charging procedure as simple as possible, Plug-less Power seeks to remove the inconvenience associated with plug-in charging, improving convenience and promoting the use of EVs [79].

IV. ON-BOARD ENERGY RESOURCES

There are seven on-board energy resources, let us discuss them in detail below.

A. LITHIUM-ion BATTERIES

EVs often run on lithium-ion batteries owing to their elevated energy density, high efficiency, and extended lifespan [28], [72], [73]. They serve as the main electrochemical component in batteries, enabling quick charging and discharging [11]. Because of their reduced weight, EVs run a longer range as compared to other energy resources [35]. However, they need to be handled carefully to prevent overheating. As this technology develops, it becomes more and more feasible for EV adoption to become mainstream.

B. NICKEL-METAL HYDRIDE (NIMH) BATTERIES

Although less frequent in contemporary EVs, NiMH battery packs stand out due to their durability and a longer lifespan than some other battery types. NiMH batteries were formerly a well-liked choice for hybrid vehicles [45]. When contrast lead-acid batteries, are more ecologically friendly and have a well-known energy density. Although NiMH batteries are usually inadequate and heavier compared to lithium-ion batteries, they are renowned for their resilience and capacity to withstand overcharging and severe temperatures. These days, its main uses are in some hybrid markets where cost is a critical factor [72]. NiMH battery may offer an alternative with benefits like endurance and affordability, as compared to lithium-ion batteries.

C. SOLID-STATE BATTERIES

They are novel innovations in this field as they employ solid electrolyte substances rather than liquid ones. Higher energy capability, quicker charging periods, and increased safety owing to fewer leaks and fire hazards are just a few benefits of this design [67]. It is anticipated that solid-state batteries would greatly increase the range of EVs and shorten their charging periods. They still need to overcome issues with scalability and manufacturing costs before they can be made commercially viable [74].

D. LEAD-ACID BATTERIES

It is one of the first variants of rechargeable batteries, which is still widely used in conventional automotive applications but is less prevalent in contemporary EVs because of its high weight and low energy density [67], [71]. They are utilized in certain EVs for auxiliary purposes including lighting, ignition, and starting despite these drawbacks. These batteries are cheap and produce a lot of power; however, they are less suitable for basic propulsion than lithium ion because of their shorter battery life and maintenance needs [67], [74].

E. ULTRA-CAPACITORS

Super-capacitors, often referred to as ultra-capacitors, are used in EVs to supplement conventional batteries by delivering brief energy bursts. Because of their ability to charge and discharge quickly and their ability to store energy electro-statically, they are perfect for operations that need high energy for brief stages, for instance, regenerative brakes and acceleration [72]. Ultra-capacitors possess less dense energy than batteries, which restricts their usage as the main source of power even if they have a longer cycle life and good power density. Rather, they are frequently used with battery packs to improve efficiency and performance [65], [67].

F. HYDROGEN FUEL CELLS

Hydrogen and oxygen combine chemically to develop just heat and water as a byproducts, which is how hydrogen fuel cells produce energy. Similar to traditional gasoline vehicles, this sustainable energy source has an elevated energy density and enables quick recharging [72]. Because of its effectiveness and advantages for the environment, fuel cells made from hydrogen are beneficial for longer-distance travel and heavier vehicles. Widespread adoption is hampered by the fact that the infrastructure needed for hydrogen generation, storage, and delivery is still in the development stages [20].

G. SOLAR PANELS

Solar panels are helpful for supplemental charging, particularly in isolated or off-grid areas, even if they normally cannot provide enough electricity to function as the main energy source [18], [72]. By extending EVs' range and lowering their reliance on grid power, solar panel integration with EVs might encourage the expansion of renewable and environmentally friendly energy sources' applications in transportation. There are many opportunities for EV applications as solar technology becomes more viable and efficient [67].

Finally, on-board energy sources are summarized in Table 4. Lithium-ion batteries outperform others in specific energy, energy density, and cycle life, but have moderate

Sources	Main Features	Drawbacks	Usage
Lithium-ion	High energy density, long lifespan,	Requires careful management	To avoid primary
Battery	efficient, rapid charging/discharging	Requires careful management	power source
Nickel-Metal	Reliable, longer lifespan, environmental	Reduced energy density and efficiency,	Used in hybrid
Battery	friendly compared to lead-acid	substantial than lithium-ion batteries	vehicles
Solid-State	Enhanced energy compactness, rapid	High manufacturing costs, scalability	Emerging
Battery	charging, time frame, elevated safety	issues	technology
Lead-Acid	Inexpensive, high power output, widely	Low energy density, heavy, shorter	Auxiliary
Battery	available	lifespan, maintenance required	functions
Ultra-	Rapid charging/discharging, extensive	Low energy density and not suitable	Complementary to
capacitor	power compactness, extended lifespan	as alone primary power source	battery systems.
Hydrogen	Exceptional energy compactness, fast	Infrastructure for hydrogen production	Long-range and
Fuel Cell	refueling, clean emissions (water heat)	storage is under developed, high costs	heavy-duty EVs.
Solar Panel	Renewable energy, reduces dependency	Insufficient as the sole power source as it	Auxiliary charging,
Solai rallel	on grid electricity, can extend EV range	is dependant on available sunlight	remote/off-grid

TABLE 4. Summary of on-board energy sources for EV charging [11], [28], [35], [45], [67], [72].

costs. On the other hand, lead-acid batteries are less expensive but lag in energy metrics and cycle life. Furthermore, it is also evident that Sodium-Nickel Chloride and NiMH batteries offer balanced trade-offs for various applications and exhibit moderate performance across all criteria [80].

V. OFF-BOARD ENERGY RESOURCES

The seven off-board EV energy sources covered in this section are:

A. GRID ELECTRICITY

The most popular energy source for charging electric automobiles is grid electricity. It includes using the electrical grid's AC, mostly through installed Level 1 and Level 2 outlets in public spaces, offices, and households. Because it leverages the electrical infrastructure that is currently in place, this approach is both broadly available and practical for EV owners. Controlling charging periods, such as postponing home charging, might lessen the effects of peak demand on the grid [81].

B. SOLAR PANELS

By turning sunshine into energy, solar panels provide an environmentally friendly power source for EVs. This renewable energy source may be utilized directly or indirectly through energy storage devices to charge EVs in public, commercial, and residential settings. Solar-powered charging stations contribute to a greener energy ecosystem by minimizing the utilization of petroleum-based fuels and emitting GHG wastage. When paired with battery storage systems, they work to provide reliable power availability [55], [82].

C. WIND TURBINES

EVs may be charged by wind energy, which is captured by wind turbines and utilized to create electricity. This renewable energy source may be used in off-grid configurations with storage options or incorporated into the grid [55]. This is an environmentally friendly and sustainable way to lessen the GHG emissions. To provide a steady energy supply for EVs,

wind power fluctuation can be controlled by integrating it with other renewable sources as well as battery storage [83].

D. HYDROPOWER PLANTS

Electricity is produced by hydropower facilities using the flow of water, usually through rivers or dams. EVs may be charged by this renewable energy source, which offers a steady and dependable supply of electricity. The consistent and steady flow of electricity produced by hydropower facilities is crucial for preserving grid stability. This is essential for EV charging, particularly when the number of EVs rises and puts more strain on the electrical networks [85]. Moreover, Hydropower facilities are perfect for supporting extensive EV charging networks since they can produce steady power and assist in controlling peak loads. Hydropower integration with EV charging can improve the environmental sustainability of transportation networks [83].

E. DIESEL GENERATORS

n isolated or off-grid areas, diesel generators are employed for an alternative power source for recharge EVs. They are not eco-friendly, but they offer a dependable power source in the absence of other options. Diesel generators may be extremely useful in emergencies or places without enough grid infrastructure [82]. But because of their expensive fuel and significant pollutants, they are usually only used seldom and are not meant to be a main source of power. Instead, they are seen as a temporary or supplemental alternative [81].

F. FUEL CELLS

In fuel cells, the sole outcomes of the chemical interaction between oxygen and hydrogen are heat and water which allows them to produce electricity. When EVs are charged using this clean energy source, it offers a cost-effective and sustainable substitute for traditional power sources [55]. Hydrogen fuel cells are appropriate for use in both consumer and electrically-powered vehicles due to their substantial amount of energy and rapid refilling periods. Additionally, they are in favor of switching to a low-carbon power system [85].

Source	Main Features	Capacity	Share (%)	Impact	Cost	Key Features
Grid	Widely available	Depends on	High	Varies by	Variable;	Most common source
Electricity	varies by region	grid		source	usually lower	dependent on local grid
Solar Panels	Renewable	1 kW to	Growing	Low	High initial	Intermittent benefits from
		500 MW			lower O&M	subsidies and incentives
Wind	Renewable	1 MW to 3 MW	Growing	Low	High initial	Intermittent suitable
Turbines		per turbine			moderate O&M	for windy areas
Hydropower	Renewable and	10 kW to	Significant	Low	High initial	Impact varies with scale
Plants	steady output	10+ MW			low O&M	can affect ecosystems
Diesel	Reliable	10 kW	Declining	High	Moderate to	Mostly used for backup
Generators		to 2 MW			high	high emissions
Fuel Cells	Clean and	1 kW to	Emerging	Low	High initial	Hydrogen production
	high efficiency	several MWs			lower O&M	depend on source
Battery Stor-	Stores energy	1 kWh to	Growing	Low	High initial	Useful for balancing
age Systems	and easily scalable	100+ MWh			moderate O&M	intermittent sources

TABLE 5. Summary of off-board energy sources for EV charging [55], [80], [81], [82], [83], [85].

G. BATTERY STORAGE SYSTEMS

To be utilized for charging EVs, battery packs store energy from numerous sources such as the electricity grid, wind turbines, solar panels, and other renewable. Through the process of storing excessive power during shortages and leaving it considering excessive need, thus these devices improve grid stability [84], [85]. They make it possible to use renewable energy more effectively, which lessens dependency on fossil fuels and increases the capacity of EV infrastructure for overall sustainability. Battery storage is crucial for integrating energy from sustainable resources into the transportation sector [82].

Finally, the summary of energy diversification on off-board EV power sources is illustrated in Table 5. Liquid fuels, gaseous fuels, electricity, or hydrogen can be produced from energy resources such as oil, natural gas, biomass, coal, hydro, wind, solar, oceanic, geothermal, and nuclear. Moreover, the range of energy carriers and channels utilized to move towards cleaner vehicle technologies, such as electric and hydrogen-powered systems, is highlighted.

VI. EV CHARGING STANDARDS AND REGULATIONS

The EV charging standards and its regulation. We will focus on (i) charging technology standards, (ii) coupler/connector standards, and (iii) communication standards.

A. CHARGING TECHNOLOGY STANDARDS

To ensure that EVs operate efficiently in the coming decades, new charging protocols, global standards, and adequate infrastructure are required. There is a correlation between the price of the infrastructure necessary for the charging points and the hardware needs. There is a chance that some EV charging standards and regulations might make the charging infrastructure substantially more expensive and intricate than the electrical infrastructure that exists now [16]. Several international organizations are working to develop standards and regulations to facilitate the charging of EVs, for example, the IEEE, IEC, SAE, ISO, etc. [86]. Moreover, the general classification is described below and summarised in Table 6.

- · Conductive Charging: The defined procedures and technical requirements that control the electrical and physical connections made directly between EVs and charging stations. These guidelines guarantee efficiency, safety, and compatibility in various EV models and charging systems. The two most well-known conductive charging standards are Type 1 and Type 2. The CCS, which combines Type 1 and Type 2 connections to enable AC and high-power DC charging over a single interface and foster interoperability, is another essential standard. These standards includes power ratings, connection design, communication protocols, and safety features that protect against current and heating issues [47]. To ensure access to an extensive selection of charging stations, conductive charging standards are fundamental elements to the seamless functioning of the entire EV ecosystem. This facilitates greater adoption and user comfort. Future developments can focus on improving power delivery capacities and including increasingly advanced safety and communication features to satisfy the growing need for fast charging facilities [19], [32], [68].
- Inductive Charging: Electromagnetic fields are used to transport energy from the on-ground charging component to the EV's. By eliminating the need for physical connectors, this technology improves convenience and reduces the wear and tear associated with conductive charging [87]. For inductive charging, the SAE J2954 standard is of vital importance due to its extensive procedures for WPT, of as much as 11kW, with plans to enable higher power levels in future expansions. Resonant inductive coupling, which transmits power via an air gap to ensure safe and effective energy transmission, is how the system functions. Several user convenience concerns are addressed by inductive charging, especially for those with physical restrictions or inclement weather [88]. There are still issues to be resolved, such as how to transport energy efficiently across different distances and how much it will cost to construct infrastructure. In order to maximize energy

Category/Code	Scope	Key Aspects				
Conductive Charging Standard						
SAE J1772	Specifies requirements for WPT	Enhances user convenience by minimising the requirement for				
	up to 11 kW	physical connectors, supports safety measures and alignment				
IEC 61851	Standard specifications for	Covers different modes of charging (Mode 1 to Mode 4), addresses				
	conductive charging systems	safety requirements and provides guidelines for communications				
COC GB/T 18487	Specifies the basic requirements	Covers both AC and DC charging, defines communication				
		protocols to ensures interoperability within the region				
JIS D 61851	Specifies the basic requirements	Ensures interoperability, covers both AC and DC charging, aligns				
		with international standards like IEC 61851				
IEC 61851-1	Outlines fundamental specifications	Covers functional requirements and testing procedures				
IEC 61851-23	Specifies requirements for DC	Focuses on DC fast charging, and procedures for interaction				
IEC 61851-24	Control communication protocol	Ensures reliable communication for efficient charging				
IEC 62196-3	Specifies requirements for DC	Includes Combo 1 as well as Combo 2 connectors, supports				
		high-power DC fast charging				
		Charging Standards				
SAE J1773	Specifies requirements for WPT	Enhances user convenience by removing the demand for physical				
	up to 11 kW	connectors, supports security measures and alignment guidelines				
SAE J2954	Specifies requirement for WPT	To ensure compatibility and interoperability between different				
	from 3.7 kW to 11 kW	manufacturers' and systems				

TABLE 6. EV charging technology standard [19], [47], [68].

TABLE 7. EV coupler/connector standards [16], [19], [47], [68], [87], [88].

Code	Scope, Charging stations	Key Aspects
IEC 62196-1	Requirement for plug, socket-outlet, & coupler	Covers mechanical, and electrical performance
IEC 62196-2	Dimensional requirements for compatibility	Specifications for Type 1 and Type 2 connector
IEC 62196-3	Requirements for fast DC charging	Covers design, performance, and safety (e.g., CCS)
SAE J1772	Physical and electrical specifications for AC	Commonly known as Type 1 connector
SAE J3068	Requirements for AC and DC connectors	Covers the Type 2 and 3 (Mennekes) connectors
SAE J1772-	Revision of SAE J1772 with updates	Includes improvements and clarifications to connector
2009		specifications for better performance and safety
ISO 15118	Defines communication protocols	Digital connectivity for grid integration
ISO 61851-1	Defines conductive charging system	Includes connectors and couplers
UL 2594	Covers safety standards for components	Includes requirements to ensure safe operation
UL 2251	Specifies safety and performance requirements	Focuses on safety, durability, and protection
DIN 70121	Communication protocols in Germany	Communication standards for control and monitoring
GB/T 20234	Covers connectors and couplers in China	Specifications for AC and DC connectors

TABLE 8. EV communication standards [86], [89].

Code	Description	Communication Protocol	Key Aspects
ISO 15118	V2G communication	Between EV and charging station	Smart charging, and grid integration
IEC 61851-1	Conductive charging	EV charging and communications	Control and data exchange
ISO 14230	Keyword 2000	Used for diagnostics	EV diagnostics
ISO 9141	Diagnostic systems	On-board diagnostics	Protocol for physical and data link
ISO 15765	CAN diagnostics	CAN networks	Protocols for diagnostics communication
SAE J1939	Truck and bus	Heavy duty vehicle networks	Diagnostics and vehicle control
SAE J1772	EV and PHEV	AC charging	Protocols for vehicle-to-AC charger
OCP	Open charge point	EV chargers and central systems	Interoperability and remote management
GB/T 27930	National standard	Charging in China	AC and DC charging standards
DIN 70121	Control and monitoring	Charging in Germany	DC charging
IEC 62056	Meters and tariff	Smart meters and data exchange	Smart grid communication and metering

utilization and encourage wider adoption of EVs, future developments are probably going to concentrate on standardizing greater power levels, enhancing synchronization and effectiveness, and interfacing with smart grid [19], [68].

B. COUPLER/CONNECTOR STANDARDS

EVs must adhere to coupler and connection standards in order to be compatible, safe, and effective. By defining the electrical and mechanical requirements for the connections that connect EVs to charging stations, these standards provide dependable and consistent charging experiences for EVs from various geographic locations. Among the significant standards are Type 1, which is utilized extensively in North America and Japan and allows AC charging as high as 19.2kW, and Type 2 which is prevalent in Europe and can handle DC as well as AC charging up to 350kW [47]. High-power DC rapid charging is supported by the CCS, It integrates Type 1 as well as Type 2 connection functionality to improve convenience and compatibility. Another important standard that is extensively utilized in Japan is CHAdeMO, which allows for high-power DC charging. To avoid dangers like overheating and electrical malfunctions, these standards address important topics such as connection architecture, security precautions, communication protocols, and energy ratings. Applications in the real world, like the Supercharger network and Tesla's unique connector, highlight how crucial it is to follow these guidelines for quick and easy charging [87], [88]. Future developments can concentrate on delivering more power, enhancing safety features, and using smart technology to expedite the charging procedure further. Table 7 displays the summary of the EV coupler and connector standards employed by the industry.

C. COMMUNICATION STANDARDS

The interoperability and effective operation of EVs within the larger transportation ecosystem depend heavily on EV communication protocols. To ensure smooth integration and optimal performance, these standards regulate the data transmission protocols between EVs and different external organizations, including grid operators, charging stations, and other cars. Important standards include ISO 15118, which outlines the communication interface between EVs and infrastructure. This allows appropriate functionalities including plug and charge, which automates payment and authentication to improve customer convenience [89]. The IEEE 802.11p standard is also essential since it is a component of the DSRC protocol, which allows V2X transmission for enhanced safety and traffic management. A variety of EV networks are supported by the OCPP, which makes it possible for charging stations and central management systems to communicate. Implementations in the real world, like Audi's usage of V2X technology, show how beneficial these standards are for enhancing traffic flow and lowering accident rates. The integration of 5G technology to enable high-speed, low-latency communication and the addition of sophisticated cyber security measures to guard against data breaches and guarantee the reliability of systems for communication are expected to be the main areas of future development for EV communication standards [86]. Table 8 illustrates the comprehensive summary of EV Communication Standards.

VII. DISCUSSION

As EVs continue to gain popularity in the transportation sector, there are advantages and disadvantages to their integration. This discourse will put our results in the broader context of the EV ecosystem and examine how they could affect safety regulations, grid stability, the growth of the infrastructure for charging EVs, and future research initiatives. The following are the key findings and concerns:

• Development of EV Charging Infrastructure: A commensurate increase of the infrastructure for charging EVs is necessary due to their increasing spread. Our investigation classifies the level of EV charging methods at the moment, making a distinction between off-board and on-board systems, and emphasizes the necessity of infrastructure that can accommodate a range of charging needs [90]. According to this review, fastcharging stations are essential for shortening charging periods and enhancing consumer convenience, but they also put a heavy load on the grid. Research into more effective and grid-friendly charging options, like usage of renewable power sources, is therefore becoming more and more crucial. In order to avoid overloading certain grid segments and to guarantee fair access to charging stations in rural as well as urban regions, the geographic placement of charging stations must also be meticulously planned. To support effective infrastructure planning, Table 9 summarizes the key parameters that must be considered when designing an EV charging system. These parameters-ranging from power and safety requirements to grid impact and user interface-are critical to ensuring reliable, scalable, and user-centric deployment of charging solutions. Understanding and controlling these factors is essential for developing a resilient and future-ready EV charging infrastructure.

- EV Safety Standards and Regulations: Safety is the foremost concern in the design and implementation of EV charging stations [91], [92], [93]. The study outlines a number of worldwide safety guidelines and regulations that regulate EV charging, but it also points out weaknesses that must be fixed to guarantee these systems operate safely. New safety challenges arise with the evolution of technology, especially with regard to high-power lightning-fast charging and the incorporation of V2G systems. The development of sophisticated diagnostic instruments for early defect detection, better thermal control mechanisms, and strong cybersecurity protocols to guard against potential attacks are just a few of the safety measures that will need to be improved in the future [94].
- Grid Stability and Integration Challenges: The effect of widespread EV integration on grid stability is one of the most significant concerns raised [95]. Due to EV charging, there is a rise in power demand, which puts the system at danger of voltage fluctuations, higher load, and even outages. According to our analysis, innovative control techniques like smart charging and, V2G solutions are essential for reducing these hazards. By adjusting charging schedules in response to grid conditions, these techniques can lower peak demand and improve grid resilience. However, such technologies necessitate collaboration amongst stakeholders, particularly grid operators, legislators, and EV manufacturers.
- Hydrogen Vehicles: The automotive industry is undergoing a notable transition as electric and hydrogen cars become the front-runners in the competition for environmentally friendly mobility. This shift presents a difficult

Category	Parameters	Importance	Design Impact	Control Strategy
Power Requirements	Charging Power, Voltage Level, Current Rating	***	Performance, Compatibility	Power electronics, transformer design, current sensors
Charging Mode	AC/DC, Conductive/Inductive, Level 1/2/3, Mode 1–4	***	Infrastructure, Efficiency	Charging topology selection, system architecture
Battery Compatibility	Battery Type, Capacity, SoC	***	Battery Safety, Efficiency	BMS integration, charge control algorithms
Communication Pro- tocols	ISO 15118, OCPP, IEC 61851, CAN	**	Interoperability, Control	Communication modules, firmware
Safety Measures	Grounding, Overcurrent, Thermal Monitoring, Isolation	***	Safety, Regulation	Fuses, circuit breakers, temperature and voltage sensors
Grid Impact	Load Management, V2G, Power Factor, Demand Response	**	Grid Stability	Smart charging, dynamic scheduling
Connector Type	CCS, CHAdeMO, Type-1/2, Tesla NACS	**	Compatibility	Hardware selection, adapter support
Environmental Factors	Temperature, IP Rating, Humidity	**	Reliability	Enclosure design, derating factors
User Interface	Display, Payment Options, Authentication	*	Usability, Adoption	HMI design, mobile app integration
Installation Site	Urban/Rural, Residential/Commercial	**	Cost, Permitting	Site survey, modular layout, compliance planning

TABLE 9. Key parameters for designing an EV charging system. Importance levels: * - Low	w, ★★ - Medium, ★★★ - high [14], [20], [34], [41], [86], [90], [95].
---	--

task because each technology has its own set of benefits and drawbacks. Battery-powered EVs have become more popular because of their increased efficiency, less emissions, and expanding infrastructure for charging. They are hampered, nevertheless, by their short range, lengthy intervals for charging, and the environmental effects of battery manufacturing and disposal. However, due to the high cost of producing hydrogen, the scarcity of refuelling places, and the requirement for additional infrastructure, hydrogen vehicles fail to live up to their promises of quicker refilling and longer ranges. The battle between electric and hydrogen-powered cars will determine the future direction of the automotive industry, impacting customer acceptance, technological advancement, and international environmental objectives as manufacturers and legislators work through these obstacles.

A. FUTURE RESEARCH DIRECTIONS

The synthesis of the literature identifies several crucial areas that require more investigation, as discussed below:

- Infrastructure and Grid Integration: More thorough research is a vital need to determine how EV charging affects grid infrastructure over the long run, especially in situations where EV penetration is significant. Most specifically, smart grid integration, real-time load balancing, and the strategic placement of charging stations to support both urban and rural adoption.
- Technological Advancements: Further research is essential to develop more economical and efficient EV charging technologies. This contains advancements in bi-directional charging (V2G), solid-state batteries, and high-power fast charging solutions that uphold battery health and safety.
- VOLUME 13, 2025

- Sustainability and Renewable Energy Integration: To build more environmentally sustainable energy ecosystems. The extensive research should explore the integration of renewable sources such as solar and wind into EV charging infrastructures. That can be supported by scalable battery management systems (BMS) for reliable energy supply.
- Interdisciplinary Collaboration and Policy Research: The complexity of EV integration calls for interdisciplinary collaboration among experts in computer science, electrical engineering, materials science, and policy. Such efforts are essential to develop holistic solutions that address technical, economic, and regulatory challenges, ensuring widespread societal and industrial benefits.

VIII. CONCLUSION

With a focus on advancements from 2020 to 2025, this review has offered a thorough overview of the state of EV charging methods, coupler technologies, energy sources, and charging standards that shape the current EV ecosystem. By synthesizing insights from 81 pivotal studies, the paper offers a cohesive and up-to-date perspective on the evolving EV charging landscape, making it a valuable reference for both researchers and industry stakeholders. The paper also suggests a number of directions for future study, stressing the significance of creating charging methods that are more sustainable and efficient, investigating the incorporation of energy sources that are renewable, and encouraging interdisciplinary cooperation. To fully achieve EVs' potential as an essential component of the global power revolution, these difficulties must be addressed. By offering a structured and comprehensive synthesis of recent advancements, this paper serves as a practical roadmap for developing efficient, scalable, and sustainable EV charging solutions. In summary,

even though EV charging methods have come a long way, more research and development are still needed to get beyond the remaining challenges. By achieving this, the EV sector will be better positioned to support the worldwide transition to more environmentally friendly and sustainable modes of transportation, which will ultimately help achieve more ambitious energy and environmental goals.

REFERENCES

- C. Johnston and E. Sobey, "How did we get here? A history of EVs," in *The Arrival of the Electric Car: Buyer's Guide, Owner's Guide, History, Future.* Warrendale, PA, USA: SAE, 2023, pp. 71–84.
- [2] J. Park, H. Lim, S. Kim, J. Song, and Y. K. Lee, "Modeling and validation of driving performance of electric vehicle converted from internal combustion engine vehicle," *IEEE Trans. Transport. Electrific.*, vol. 11, no. 1, pp. 486–498, Feb. 2025, doi: 10.1109/TTE.2024.3392208.
- [3] A. P. Cohen, S. A. Shaheen, and E. M. Farrar, "Urban air mobility: History, ecosystem, market potential, and challenges," *IEEE Trans. Intell. Transp. Syst.*, vol. 22, no. 9, pp. 6074–6087, Sep. 2021, doi: 10.1109/TITS.2021.3082767.
- [4] D. Said, "A survey on information communication technologies in modern demand-side management for smart grids: Challenges, solutions, and opportunities," *IEEE Eng. Manag. Rev.*, vol. 51, no. 1, pp. 76–107, Mar. 2023, doi: 10.1109/EMR.2022.3186154.
- [5] W. B. Abbas, Q. Z. Ahmed, F. A. Khan, N. S. Mian, P. I. Lazaridis, and P. Sureephong, "Designing future wireless networks (FWN)s with net zero (NZ) and zero touch (ZT) perspective," *IEEE Access*, vol. 11, pp. 83301–83321, 2023, doi: 10.1109/ACCESS.2023.3301849.
- [6] Environmental Protection Agency. (2022). Sources of Greenhouse Gas Emissions. EPA. [Online]. Available: https://www.epa.gov/ghgemissions/ sources-greenhouse-gas-emissions
- [7] M. Asna, H. Shareef, A. Prasanthi, R. Errouissi, and A. Wahyudie, "A novel multi-level charging strategy for electric vehicles to enhance customer charging experience and station utilization," *IEEE Trans. Intell. Transp. Syst.*, vol. 25, no. 9, pp. 11497–11508, Sep. 2024, doi: 10.1109/TITS.2024.3372183.
- [8] A. Ahmad, M. S. Alam, and R. Chabaan, "A comprehensive review of wireless charging technologies for electric vehicles," *IEEE Trans. Transport. Electrific.*, vol. 4, no. 1, pp. 38–63, Mar. 2018, doi: 10.1109/TTE.2017.2771619.
- [9] H. Tu, H. Feng, S. Srdic, and S. Lukic, "Extreme fast charging of electric vehicles: A technology overview," *IEEE Trans. Transport. Electrific.*, vol. 5, no. 4, pp. 861–878, Dec. 2019, doi: 10.1109/TTE.2019.2958709.
- [10] A. Khaligh and M. D'Antonio, "Global trends in high-power on-board chargers for electric vehicles," *IEEE Trans. Veh. Technol.*, vol. 68, no. 4, pp. 3306–3324, Apr. 2019, doi: 10.1109/TVT.2019.2897050.
- [11] O. M. F. Camacho, P. B. Nørgård, N. Rao, and L. Mihet-Popa, "Electrical vehicle batteries testing in a distribution network using sustainable energy," *IEEE Trans. Smart Grid*, vol. 5, no. 2, pp. 1033–1042, Mar. 2014, doi: 10.1109/TSG.2014.2299064.
- [12] A. Thingvad, L. Calearo, P. B. Andersen, and M. Marinelli, "Empirical capacity measurements of electric vehicles subject to battery degradation from V2G services," *IEEE Trans. Veh. Technol.*, vol. 70, no. 8, pp. 7547–7557, Aug. 2021, doi: 10.1109/TVT.2021.3093161.
- [13] E. S. Rigas, S. D. Ramchurn, and N. Bassiliades, "Managing electric vehicles in the smart grid using artificial intelligence: A survey," *IEEE Trans. Intell. Transp. Syst.*, vol. 16, no. 4, pp. 1619–1635, Aug. 2015, doi: 10.1109/TITS.2014.2376873.
- [14] M. Yilmaz and P. T. Krein, "Review of battery charger topologies, charging power levels, and infrastructure for plug-in electric and hybrid vehicles," *IEEE Trans. Power Electron.*, vol. 28, no. 5, pp. 2151–2169, May 2013, doi: 10.1109/TPEL.2012.2212917.
- [15] L. Wang, Z. Qin, T. Slangen, P. Bauer, and T. van Wijk, "Grid impact of electric vehicle fast charging stations: Trends, standards, issues and mitigation measures—An overview," *IEEE Open J. Power Electron.*, vol. 2, pp. 56–74, 2021, doi: 10.1109/OJPEL.2021.3054601.
- [16] M. R. Khalid, I. A. Khan, S. Hameed, M. S. J. Asghar, and J. Ro, "A comprehensive review on structural topologies, power levels, energy storage systems, and standards for electric vehicle charging stations and their impacts on grid," *IEEE Access*, vol. 9, pp. 128069–128094, 2021, doi: 10.1109/ACCESS.2021.3112189.

- [17] M.-F. Ng, J. Zhao, Q. Yan, G. J. Conduit, and Z. W. Seh, "Predicting the state of charge and health of batteries using data-driven machine learning," *Nature Mach. Intell.*, vol. 2, no. 3, pp. 161–170, Mar. 2020, doi: 10.1038/s42256-020-0156-7.
- [18] S. A. Q. Mohammed and J.-W. Jung, "A comprehensive stateof-the-art review of wired/wireless charging technologies for battery electric vehicles: Classification/common topologies/future research issues," *IEEE Access*, vol. 9, pp. 19572–19585, 2021, doi: 10.1109/ACCESS.2021.3055027.
- [19] S. S. G. Acharige, M. E. Haque, M. T. Arif, N. Hosseinzadeh, K. N. Hasan, and A. M. T. Oo, "Review of electric vehicle charging technologies, standards, architectures, and converter configurations," *IEEE Access*, vol. 11, pp. 41218–41255, 2023, doi: 10.1109/ACCESS.2023.3267164.
- [20] D. Ronanki and H. Karneddi, "Electric vehicle charging infrastructure: Review, cyber security considerations, potential impacts, countermeasures, and future trends," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 12, no. 1, pp. 242–256, Feb. 2024, doi: 10.1109/JESTPE.2023.3336997.
- [21] T. Yuvaraj, K. R. Devabalaji, J. A. Kumar, S. B. Thanikanti, and N. I. Nwulu, "A comprehensive review and analysis of the allocation of electric vehicle charging stations in distribution networks," *IEEE Access*, vol. 12, pp. 5404–5461, 2024, doi: 10.1109/ACCESS.2023.3349274.
- [22] H. Ko, S. Pack, and V. C. M. Leung, "An optimal battery charging algorithm in electric vehicle-assisted battery swapping environments," *IEEE Trans. Intell. Transp. Syst.*, vol. 23, no. 5, pp. 3985–3994, May 2022, doi: 10.1109/TITS.2020.3038274.
- [23] A. Ahmad, Z. Qin, T. Wijekoon, and P. Bauer, "An overview on medium voltage grid integration of ultra-fast charging stations: Current status and future trends," *IEEE Open J. Ind. Electron. Soc.*, vol. 3, pp. 420–447, Jan. 2022, doi: 10.1109/OJIES.2022.3179743.
- [24] O. N. Nezamuddin, C. L. Nicholas, and E. C. d. Santos, "The problem of electric vehicle charging: State-of-the-art and an innovative solution," *IEEE Trans. Intell. Transp. Syst.*, vol. 23, no. 5, pp. 4663–4673, May 2022, doi: 10.1109/TITS.2020.3048728.
- [25] J. Yuan, L. Dorn-Gomba, A. D. Callegaro, J. Reimers, and A. Emadi, "A review of bidirectional on-board chargers for electric vehicles," *IEEE Access*, vol. 9, pp. 51501–51518, 2021, doi: 10.1109/ACCESS.2021.3069448.
- [26] J. Li, R. Qin, C. Olaverri-Monreal, R. Prodan, and F.-Y. Wang, "Logistics 5.0: From intelligent networks to sustainable ecosystems," *IEEE Trans. Intell. Vehicles*, vol. 8, no. 7, pp. 3771–3774, Jul. 2023, doi: 10.1109/TIV.2023.3295796.
- [27] S. Semsar, T. Soong, and P. W. Lehn, "Integrated single-phase electric vehicle charging using a dual-inverter drive," in *Proc. IEEE Transp. Electrific. Conf. Expo (ITEC)*, Long Beach, CA, USA, Jun. 2018, pp. 320–325, doi: 10.1109/ITEC.2018.8450132.
- [28] A. Mahesh, B. Chokkalingam, and L. Mihet-Popa, "Inductive wireless power transfer charging for electric vehicles—A review," *IEEE Access*, vol. 9, pp. 137667–137713, 2021, doi: 10.1109/ACCESS.2021.3116678.
- [29] S. Padmanaban, T. Samavat, M. A. Nasab, M. A. Nasab, M. Zand, and F. Nikokar, "Electric vehicles and IoT in smart cities," in *Artificial Intelligence-Based Smart Power Systems*. Wiley, 2023, ch. 14, pp. 273–290, doi: 10.1002/9781119893998 ch14.
- [30] M. Safayatullah, M. T. Elrais, S. Ghosh, R. Rezaii, and I. Batarseh, "A comprehensive review of power converter topologies and control methods for electric vehicle fast charging applications," *IEEE Access*, vol. 10, pp. 40753–40793, 2022, doi: 10.1109/ACCESS.2022.3166935.
- [31] J. Levinson, J. Askeland, J. Becker, J. Dolson, D. Held, S. Kammel, J. Z. Kolter, D. Langer, O. Pink, V. Pratt, M. Sokolsky, G. Stanek, D. Stavens, A. Teichman, M. Werling, and S. Thrun, "Towards fully autonomous driving: Systems and algorithms," in *Proc. IEEE Intell. Vehicles Symp. (IV)*, Jun. 2011, pp. 163–168, doi: 10.1109/IVS.2011.5940562.
- [32] Y. Liu, Y. Zhu, and Y. Cui, "Challenges and opportunities towards fastcharging battery materials," *Nature Energy*, vol. 4, no. 7, pp. 540–550, Jun. 2019, doi: 10.1038/s41560-019-0405-3.
- [33] X. Chen, Z. Li, H. Dong, Z. Hu, and C. C. Mi, "Enabling extreme fast charging technology for electric vehicles," *IEEE Trans. Intell. Transp. Syst.*, vol. 22, no. 1, pp. 466–470, Jan. 2021, doi: 10.1109/TITS.2020.3045241.
- [34] M. E. Kabir, C. Assi, H. Alameddine, J. Antoun, and J. Yan, "Demandaware provisioning of electric vehicles fast charging infrastructure," *IEEE Trans. Veh. Technol.*, vol. 69, no. 7, pp. 6952–6963, Jul. 2020, doi: 10.1109/TVT.2020.2993509.

- [35] C. Chen, Z. Wei, and A. C. Knoll, "Charging optimization for Li-ion battery in electric vehicles: A review," *IEEE Trans. Transport. Electrific.*, vol. 8, no. 3, pp. 3068–3089, Sep. 2022, doi: 10.1109/TTE.2021.3135525.
- [36] D. Zinchenko, A. Blinov, A. Chub, D. Vinnikov, I. Verbytskyi, and S. Bayhan, "High-efficiency single-stage on-board charger for electrical vehicles," *IEEE Trans. Veh. Technol.*, vol. 70, no. 12, pp. 12581–12592, Dec. 2021, doi: 10.1109/TVT.2021.3118392.
- [37] A. Bindra, "Electric vehicle batteries eye solid-state technology: Prototypes promise lower cost, faster charging, and greater safety," *IEEE Power Electron. Mag.*, vol. 7, no. 1, pp. 16–19, Mar. 2020, doi: 10.1109/MPEL.2019.2961203.
- [38] Y. Wang, H. T. Luan, Z. Su, N. Zhang, and A. Benslimane, "A secure and efficient wireless charging scheme for electric vehicles in vehicular energy networks," *IEEE Trans. Veh. Technol.*, vol. 71, no. 2, pp. 1491–1508, Feb. 2022, doi: 10.1109/TVT.2021.3131776.
- [39] I. Husain, B. Ozpineci, M. S. Islam, E. Gurpinar, G.-J. Su, W. Yu, S. Chowdhury, L. Xue, D. Rahman, and R. Sahu, "Electric drive technology trends, challenges, and opportunities for future electric vehicles," *Proc. IEEE*, vol. 109, no. 6, pp. 1039–1059, Jun. 2021, doi: 10.1109/JPROC.2020.3046112.
- [40] S. Li, S. Lu, and C. C. Mi, "Revolution of electric vehicle charging technologies accelerated by wide bandgap devices," *Proc. IEEE*, vol. 109, no. 6, pp. 985–1003, Jun. 2021, doi: 10.1109/JPROC.2021.3071977.
- [41] A. K. M. Yousuf, Z. Wang, R. Paranjape, and Y. Tang, "An indepth exploration of electric vehicle charging station infrastructure: A comprehensive review of challenges, mitigation approaches, and optimization strategies," *IEEE Access*, vol. 12, pp. 51570–51589, 2024, doi: 10.1109/ACCESS.2024.3385731.
- [42] D. Said and H. T. Mouftah, "A novel electric vehicles charging/discharging management protocol based on queuing model," *IEEE Trans. Intell. Vehicles*, vol. 5, no. 1, pp. 100–111, Mar. 2020, doi: 10.1109/TIV.2019.2955370.
- [43] J. M. González-González, A. Triviño-Cabrera, and J. A. Aguado, "Model predictive control to maximize the efficiency in EV wireless chargers," *IEEE Trans. Ind. Electron.*, vol. 69, no. 2, pp. 1244–1253, Feb. 2022, doi: 10.1109/TIE.2021.3057006.
- [44] N. Mohamed, F. Aymen, T. E. A. Alharbi, C. Z. El-Bayeh, S. Lassaad, S. S. M. Ghoneim, and U. Eicker, "A comprehensive analysis of wireless charging systems for electric vehicles," *IEEE Access*, vol. 10, pp. 43865–43881, 2022, doi: 10.1109/ACCESS.2022. 3168727.
- [45] X. Huang, A. B. Acharya, J. Meng, X. Sui, D.-I. Stroe, and R. Teodorescu, "Wireless smart battery management system for electric vehicles," in *Proc. IEEE Energy Convers. Congr. Expo.* (ECCE), Oct. 2020, pp. 5620–5625, doi: 10.1109/ECCE44975.2020. 9236279.
- [46] H. Feng, R. Tavakoli, O. C. Onar, and Z. Pantic, "Advances in high-power wireless charging systems: Overview and design considerations," *IEEE Trans. Transport. Electrific.*, vol. 6, no. 3, pp. 886–919, Sep. 2020, doi: 10.1109/TTE.2020.3012543.
- [47] X. Lu, P. Wang, D. Niyato, D. I. Kim, and Z. Han, "Wireless charging technologies: Fundamentals, standards, and network applications," *IEEE Commun. Surveys Tuts.*, vol. 18, no. 2, pp. 1413–1452, 2nd Quart., 2016, doi: 10.1109/COMST.2015.2499783.
- [48] V. Ramakrishnan, A. D. Savio and C. Balaji, N. Rajamanickam, H. Kotb, A. Elrashidi, and W. Nureldeen, "A comprehensive review on efficiency enhancement of wireless charging system for the electric vehicles applications," *IEEE Access*, vol. 12, pp. 46967–46994, 2024, doi: 10.1109/ACCESS.2024.3378303.
- [49] S. Li and C. C. Mi, "Wireless power transfer for electric vehicle applications," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 3, no. 1, pp. 4–17, Mar. 2015, doi: 10.1109/JESTPE.2014.2319453.
- [50] G. Buja, M. Bertoluzzo, and H. K. Dashora, "Lumped track layout design for dynamic wireless charging of electric vehicles," *IEEE Trans. Ind. Electron.*, vol. 63, no. 10, pp. 6631–6640, Oct. 2016, doi: 10.1109/TIE.2016.2538738.
- [51] P. Ridden. (2017). Korean Electric Vehicle Solution. New Atlas. Accessed: Oct. 17, 2024. [Online]. Available: https://newatlas.com/kaistolev-electric-vehicle/12557/
- [52] S. Jeong, Y. J. Jang, D. Kum, and M. S. Lee, "Charging automation for electric vehicles: Is a smaller battery good for the wireless charging electric vehicles?" *IEEE Trans. Autom. Sci. Eng.*, vol. 16, no. 1, pp. 486–497, Jan. 2019, doi: 10.1109/TASE.2018.2827954.

- [53] S. Zhang and J. J. Q. Yu, "Electric vehicle dynamic wireless charging system: Optimal placement and vehicle-to-grid scheduling," *IEEE Internet Things J.*, vol. 9, no. 8, pp. 6047–6057, Apr. 2022, doi: 10.1109/JIOT.2021.3109956.
- [54] Z. Hua, K. T. Chau, H. Pang, and T. Yang, "Dynamic wireless charging for electric vehicles with autonomous frequency control," *IEEE Trans. Magn.*, vol. 59, no. 11, pp. 1–5, Nov. 2023, doi: 10.1109/TMAG.2023.3293793.
- [55] P. Machura, V. De Santis, and Q. Li, "Driving range of electric vehicles charged by wireless power transfer," *IEEE Trans. Veh. Technol.*, vol. 69, no. 6, pp. 5968–5982, Jun. 2020, doi: 10.1109/TVT.2020.2984386.
- [56] A. A. S. Mohamed, C. R. Lashway, and O. Mohammed, "Modeling and feasibility analysis of quasi-dynamic WPT system for EV applications," *IEEE Trans. Transport. Electrific.*, vol. 3, no. 2, pp. 343–353, Jun. 2017, doi: 10.1109/TTE.2017.2682111.
- [57] H. N. T. Nguyen, C. Zhang, and J. Zhang, "Dynamic demand control of electric vehicles to support power grid with high penetration level of renewable energy," *IEEE Trans. Transport. Electrific.*, vol. 2, no. 1, pp. 66–75, Mar. 2016, doi: 10.1109/TTE.2016.2519821.
- [58] Y. Zheng, Z. Y. Dong, Y. Xu, K. Meng, J. H. Zhao, and J. Qiu, "Electric vehicle battery Charging/Swap stations in distribution systems: Comparison study and optimal planning," *IEEE Trans. Power Syst.*, vol. 29, no. 1, pp. 221–229, Jan. 2014, doi: 10.1109/TPWRS.2013. 2278852.
- [59] CAAM, CATARC, and Toyota. (2018). Annual Report on Automotive Industry in China (2018). Beijing, China: Social Sciences Academic Press. Accessed: Oct. 17, 2024. [Online]. Available: http://www. autostats.org.cn/ReadArticle.asp?NewsID=3134
- [60] M. Ban, J. Yu, Z. Li, D. Guo, and J. Ge, "Battery swapping: An aggressive approach to transportation electrification," *IEEE Electrific. Mag.*, vol. 7, no. 3, pp. 44–54, Sep. 2019, doi: 10.1109/MELE.2019. 2925762.
- [61] M. O. Tarar, N. U. Hassan, I. H. Naqvi, and M. Pecht, "Techno-economic framework for electric vehicle battery swapping stations," *IEEE Trans. Transport. Electrific.*, vol. 9, no. 3, pp. 4458–4473, Sep. 2023, doi: 10.1109/TTE.2023.3252169.
- [62] M. Ban, D. Guo, J. Yu, and M. Shahidehpour, "Optimal sizing of PV and battery-based energy storage in an off-grid nanogrid supplying batteries to a battery swapping station," *J. Modern Power Syst. Clean Energy*, vol. 7, no. 2, pp. 309–320, Mar. 2019, doi: 10.1007/s40565-018-0428-y.
- [63] R. Xie, W. Wei, Q. Wu, T. Ding, and S. Mei, "Optimal service pricing and charging scheduling of an electric vehicle sharing system," *IEEE Trans. Veh. Technol.*, vol. 69, no. 1, pp. 78–89, Jan. 2020, doi: 10.1109/TVT.2019.2950402.
- [64] V.-B. Vu, J. M. González-González, V. Pickert, M. Dahidah, and A. Triviño, "A hybrid charger of conductive and inductive modes for electric vehicles," *IEEE Trans. Ind. Electron.*, vol. 68, no. 12, pp. 12021–12033, Dec. 2021, doi: 10.1109/TIE.2020.3042162.
- [65] A. Khaligh and S. Dusmez, "Comprehensive topological analysis of conductive and inductive charging solutions for plug-in electric vehicles," *IEEE Trans. Veh. Technol.*, vol. 61, no. 8, pp. 3475–3489, Oct. 2012, doi: 10.1109/TVT.2012.2213104.
- [66] S. A. Ali, V. Pickert, M. A. Alharbi, H. Li, and H. Patsios, "Wheel hub active power collection unit for dynamic conductive road charging," *IEEE Trans. Transport. Electrific.*, vol. 9, no. 1, pp. 1927–1936, Mar. 2023, doi: 10.1109/TTE.2022.3182272.
- [67] M. Schmid, E. Gebauer, C. Hanzl, and C. Endisch, "Active modelbased fault diagnosis in reconfigurable battery systems," *IEEE Trans. Power Electron.*, vol. 36, no. 3, pp. 2584–2597, Mar. 2021, doi: 10.1109/TPEL.2020.3012964.
- [68] H. Bagheri, M. N.-A.-R. M. Noor-A-Rahim, Z. Liu, H. Lee, D. Pesch, K. Moessner, and P. Xiao, "5G NR-V2X: Toward connected and cooperative autonomous driving," *IEEE Commun. Standards Mag.*, vol. 5, no. 1, pp. 48–54, Mar. 2021, doi: 10.1109/MCOMSTD.001.2000069.
- [69] M. Yamauchi. (2019). The Complete Guide to Tesla Charging at Home, on the go and Autonomously. Plugless Power. Accessed: Jan. 9, 2025. [Online]. Available: https://pluglesspower.com/
- [70] N. Shinohara, "History and innovation of wireless power transfer via microwaves," *IEEE J. Microw.*, vol. 1, no. 1, pp. 218–228, Jan. 2021, doi: 10.1109/JMW.2020.3030896.
- [71] I. Llatser, T. Michalke, M. Dolgov, F. Wildschütte, and H. Fuchs, "Cooperative automated driving use cases for 5G V2X communication," in *Proc. IEEE 2nd 5G World Forum (5GWF)*, Dresden, Germany, Sep. 2019, pp. 120–125, doi: 10.1109/5GWF.2019.8911628.

- [72] H. Wei, J. Wu, C. Lv, S. Yang, H. Zhang, and K. Al-Haddad, "Intelligent EV charging control and management: From microscale battery cell to macroscale grid synergy," *IEEE Trans. Intell. Vehicles*, vol. 9, no. 2, pp. 1–21, May 2024, doi: 10.1109/TIV.2024.3432075.
- [73] M. A. Hannan, M. M. Hoque, A. Hussain, Y. Yusof, and P. J. Ker, "State-ofthe-art and energy management system of lithium-ion batteries in electric vehicle applications: Issues and recommendations," *IEEE Access*, vol. 6, pp. 19362–19378, 2018, doi: 10.1109/ACCESS.2018.2817655.
- [74] U. R. Muduli, A. R. Beig, K. Al Jaafari, J. Y. Alsawalhi, and R. K. Behera, "An improved direct torque control with battery power management of open-end winding induction motor drive for electric vehicles," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Detroit, MI, USA, Oct. 2020, pp. 6332–6337, doi: 10.1109/ECCE44975.2020.9236199.
- [75] Wireless Power Consortium. (2021). About QI. Accessed: Jan. 9, 2025. [Online]. Available: https://www.wirelesspowerconsortium. com/standards/qi-wireless-charging/
- [76] Qi System Description: Wireless Power Transfer: Low Power, Part 1: Interface Definition, Wireless Power Consortium, Washington, DC, USA, 2012, vol. 1.
- [77] B. Berman. (2024). CHAdeMO 3.0 to Harmonize Global EV Quick Charging Standards. SAE International. Accessed: Oct. 25, 2024. [Online]. Available: https://www.sae.org
- [78] WiTricity. (2021). WiTricity Halo: The Future of Wireless Charging for Electric Vehicles. Accessed: Jan. 9, 2025. [Online]. Available: https:// witricity.com/automotive/witricity-wireless-charging
- [79] Evatran. (2020). Plugless Power: Wireless Charging for Electric Vehicles. Accessed: Jan. 9, 2025. [Online]. Available: https://pluglesspower. com/pgls_press_release/plugless-power-acquires-evatran-wirelessinductive-charging-patent-portfolio
- [80] X. Sun, Z. Li, X. Wang, and C. Li, "Technology development of electric vehicles: A review," *Energies*, vol. 13, no. 1, p. 90, Dec. 2019.
- [81] S. Kumar, T. Upadhyay, and O. H. Gupta, "Power quality improvement and signal conditioning of PV array and grid interfaced off-board charger for electric vehicles with V2G and G2V capabilities," *Chin. J. Electr. Eng.*, vol. 9, no. 4, pp. 132–143, Dec. 2023, doi: 10.23919/cjee.2023.000027.
- [82] S. Harini, N. Chellammal, B. Chokkalingam, and L. Mihet-Popa, "A novel high gain dual input single output Z-quasi resonant (ZQR) DC/DC converter for off-board EV charging," *IEEE Access*, vol. 10, pp. 83350–83367, 2022, doi: 10.1109/ACCESS.2022.3195936.
- [83] M. R. V. Moghadam, R. Zhang, and R. T. B. Ma, "Distributed frequency control via randomized response of electric vehicles in power grid," *IEEE Trans. Sustain. Energy*, vol. 7, no. 1, pp. 312–324, Jan. 2016, doi: 10.1109/TSTE.2015.2494504.
- [84] K. W. E. Cheng, B. P. Divakar, H. Wu, K. Ding, and H. F. Ho, "Batterymanagement system (BMS) and SOC development for electrical vehicles," *IEEE Trans. Veh. Technol.*, vol. 60, no. 1, pp. 76–88, Jan. 2011, doi: 10.1109/TVT.2010.2089647.
- [85] X. S. Zhang, T. Yu, Z. N. Pan, B. Yang, and T. Bao, "Lifelong learning for complementary generation control of interconnected power grids with high-penetration renewables and EVs," *IEEE Trans. Power Syst.*, vol. 33, no. 4, pp. 4097–4110, Jul. 2018, doi: 10.1109/TPWRS.2017. 2767318.
- [86] B. Al-Hanahi, I. Ahmad, D. Habibi, and M. A. S. Masoum, "Charging infrastructure for commercial electric vehicles: Challenges and future works," *IEEE Access*, vol. 9, pp. 121476–121492, 2021, doi: 10.1109/ACCESS.2021.3108817.
- [87] S. Habib, M. M. Khan, F. Abbas, L. Sang, M. U. Shahid, and H. Tang, "A comprehensive study of implemented international standards, technical challenges, impacts and prospects for electric vehicles," *IEEE Access*, vol. 6, pp. 13866–13890, 2018, doi: 10.1109/ACCESS.2018. 2812303.
- [88] S. M. S. Hussain, T. S. Ustun, P. Nsonga, and I. Ali, "IEEE 1609 WAVE and IEC 61850 standard communication based integrated EV charging management in smart grids," *IEEE Trans. Veh. Technol.*, vol. 67, no. 8, pp. 7690–7697, Aug. 2018, doi: 10.1109/TVT.2018.2838018.
- [89] V. Cirimele, F. Freschi, and L. Zhao, "Critical comparative review of international standards on wireless charging for light-duty electric vehicles," *IEEE Trans. Ind. Appl.*, vol. 60, no. 5, pp. 7403–7412, Sep. 2024, doi: 10.1109/TIA.2024.3408107.
- [90] T. Chen, X.-P. Zhang, J. Wang, J. Li, C. Wu, M. Hu, and H. Bian, "A review on electric vehicle charging infrastructure development in the U.K.," *J. Modern Power Syst. Clean Energy*, vol. 8, no. 2, pp. 193–205, Mar. 2020, doi: 10.35833/MPCE.2018.000374.

- [91] C. Gong, J. Liu, Y. Han, Y. Hu, H. Yu, and R. Zeng, "Safety of electric vehicles in crash conditions: A review of hazards to occupants, regulatory activities, and technical support," *IEEE Trans. Transport. Electrific.*, vol. 8, no. 3, pp. 3870–3883, Sep. 2022, doi: 10.1109/TTE.2021.3136126.
- [92] S. Shirvani, Y. Baseri, and A. Ghorbani, "Evaluation framework for electric vehicle security risk assessment," *IEEE Trans. Intell. Transp. Syst.*, vol. 25, no. 1, pp. 33–56, Jan. 2024, doi: 10.1109/TITS.2023.3307660.
- [93] L. Guo and J. Ye, "Cyber-physical security of electric vehicles with four motor drives," *IEEE Trans. Power Electron.*, vol. 36, no. 4, pp. 4463–4477, Apr. 2021, doi: 10.1109/TPEL.2020.3025718.
- [94] J. Petit and S. E. Shladover, "Potential cyberattacks on automated vehicles," *IEEE Trans. Intell. Transp. Syst.*, vol. 16, no. 2, pp. 546–556, Apr. 2015, doi: 10.1109/TITS.2014.2342271.
- [95] N. Kumar, V. Saxena, B. Singh, and B. K. Panigrahi, "Power quality improved grid-interfaced PV-assisted onboard EV charging infrastructure for smart households consumers," *IEEE Trans. Consum. Electron.*, vol. 69, no. 4, pp. 1091–1100, Nov. 2023, doi: 10.1109/TCE.2023.3296480.



ANSIF AROOJ is currently pursuing the Ph.D. degree in engineering with the University of Huddersfield, U.K. Her academic journey spans esteemed institutions across Pakistan and the U.K. She has been a Lecturer in computer science with the University of Education, Lahore, since 2009 (currently on study leave). She has published in leading scientific journals, with impactful research in machine learning, transportation networks, the Internet of Vehicles (IoV), blockchain, and

computer-aided diagnostics. Her works appear in high-impact journals, such as IEEE Access, *Entropy, Sensors*, and *Journal of Network and Computer Applications*. Her research focuses on advanced connected technologies for electric vehicle safety, blending innovation, interdisciplinary collaboration, and real-world application. She has received recognitions, including the Engineering Excellence Scholarship (University of Huddersfield), selected as a Young Researcher at the Heidelberg Laureate Forum, and a Travel Grant from IEEE/GIST South Korea.



QASIM ZEESHAN AHMED (Member, IEEE) received the Ph.D. degree from the University of Southampton, Southampton, U.K., in 2009. He was a Lecturer, a Senior Lecturer, a Reader, and a Professor of electronic engineering with the School of Computing and Engineering, University of Huddersfield, Huddersfield, U.K., in 2017, 2018, 2020, and 2025, respectively. He is currently a Principal Investigator with the U.K. for the Erasmus+ DigiHealth-Asia Project and the

MSCA Staff Exchanges EVOLVE Project. He is a Co-investigator of the EU H2020 ETN Research MOTOR5G Project and the EU H2020 RISE Research RECOMBINE Project. His research interests include mainly ultrawide bandwidth systems, millimeter waves, device-to-device, digital health, and cooperative communications. He is a fellow of the Higher Education Academy.



MUHAMMAD SHOAIB FAROOQ (Senior Member, IEEE) was born in Lahore, Pakistan. He received the M.Sc. degree in computer science from Quaid-e-Azam University, in 1995, the M.Phil. degree in computer science from Government College University, in 2007, and the Ph.D. degree in computer science from Abdul Wali Khan University, in 2015. He is a Professor with the Department of Computer Science, University of Management and Technology (UMT), Lahore,

Pakistan. He is also an Affiliate Member of George Mason University, USA. With over 26 years of teaching experience in the field of computer science, he has made significant contributions to academia through his research and publications. He has authored several papers in peer-reviewed international journals and conferences. His research interests span a wide range of areas, including the theory of programming languages, big data, the Internet of Things (IoT), the Internet of Vehicles (IoV), machine learning, distributed systems, and computer science education. He is a member of the IEEE Systems, Man, and Cybernetics Society, reflecting his active engagement with the international research community.



NAUMAN ASLAM (Senior Member, IEEE) received the Ph.D. degree in engineering mathematics from Dalhousie University, Canada, in 2008. He is a Professor of network systems and security with the Department of Computer and Information Sciences, Northumbria University, U.K. Before joining Northumbria University, as a Senior Lecturer, in 2011, he was an Assistant Professor with Dalhousie University. He is leading the Cyber Security and Network

Systems (CyberNets) Research Group, Northumbria University. He has led an Erasmus Mundus Action-2 project titled "Sustainable Green Economies through Learning, Innovation, Networking and Knowledge Exchange" (gLINK), which is funded by European Union (worth 3.03 million Euros). He has published over 100 papers in peer-reviewed journals and conferences. His research interests cover diverse but interconnected areas related to communication networks. His current research efforts are focused at addressing problems related to wireless body area networks and the IoT, network security, QoS-aware communication in industrial wireless sensor networks, and application of artificial intelligence (AI) in communication networks.



TARIQ UMER (Senior Member, IEEE) received the master's degree in computer science from Bahauddin Zakariya University, Multan, Pakistan, in 1997, and the Ph.D. degree in communication systems from the School of Computing and Communication, Lancaster University, U.K., in 2012. He was with IT education sector, Pakistan, for more than 13 years. Since January 2007, he has been a Tenured Associate Professor with the CS Department, COMSATS University Islamabad,

Lahore Campus, Lahore. His research interests include vehicular ad-hoc networks, the Internet of Things, edge computing, data driven networks, digital twins, and telecommunication network design. He is an Active Member of Pakistan Computer Society and the Internet Society Pakistan. He served in the TPC for various international conferences. He is serving as a Reviewer for IEEE COMMUNICATIONS LETTERS, IEEE Access, IEEE SURVEY AND TUTORIAL, *IEEE Wireless Communications Magazine, IEEE Network, Computers and Electrical Engineering* (Elsevier), *Journal of Network and Computer Applications* (Elsevier), *Ad Hoc Sensor Wireless Networks, Wireless Networks*. He is an Editor of *Future Generation Computer Systems* (Elsevier) and an Associate Editor of IEEE Access.



TEMITOPE ALADE (Member, IEEE) received the B.Sc. degree in computer science from the University of Ilorin, the M.Sc. degree in mobile computing from the University of Bradford, and the Ph.D. degree in electronic engineering from the University of Kent. He is an Associate Professor of computing sciences and the Director of the M.Sc. Computing Program with the University of East Anglia (UEA). His academic journey spans teaching, research, and leadership roles with top

U.K. institutions, including Nottingham Trent University, the University of Worcester, the University of Keele, and the University of Kent. He is a fellow of the Higher Education Academy and actively contributes to curriculum development and postgraduate supervision, particularly in web technologies, mobile application development, and the Internet of Things (IoT). His research focuses on modeling and analysis of computing and communication systems, with interests spanning mobile and wireless communication (including 6G), the IoT, machine learning, mobile systems, and immersive software platforms. He has led and contributed to several EU-funded research projects, serves as a reviewer for leading IEEE, Elsevier, and Wiley journals, and sits on technical committees for major IEEE conferences. Actively involved in collaborative research and Ph.D. supervision, he welcomes opportunities to explore interdisciplinary applications of computing in areas, such as healthcare, finance, education, and smart systems.

...